Author’s response to reviews

Title: A Randomized, Placebo Controlled, Trial of Preoperative Sustained Release Betamethasone Plus Non-Controlled Intraoperative Ketorolac or Fentanyl on Pain After Diagnostic Laparoscopy or Laparoscopic Tubal Ligation [ISRCTN52633712].

Authors:

Warren P. Bagely (WBagley@mc.utmck.edu)
A Audie Smith (AASmith@mc.utmck.edu)
Jessie D. Hebert (JHebert@mc.utmck.edu)
Carolyn C. Snider (CSnider@mc.utmck.edu)
Marsha D. Piller (MPiller@mc.utmck.edu)
Gary A. Sega (Garysega@comcast.net)
Paula C. Carney (PCarney@utk.edu)
Prof Roger C. Carroll (RCCarroll@mc.utmck.edu)

Version: 5 Date: 19 Aug 2003

Revised title to include clinical trials number.